Author's response to reviews

Title: Possible pulmonary endocarditis in a 72-year-old woman with an uncorrected Tetralogy of Fallot

Authors:

Pedro A Sousa (Peter_senado2002@yahoo.com)
Walter Santos (Peter_senado2002@yahoo.com)
Nuno Marques (Peter_senado2002@yahoo.com)
Pedro Cordeiro (Peter_senado2002@yahoo.com)
Rui Ferrinha (Peter_senado2002@yahoo.com)
Salomé Pereira (Peter_senado2002@yahoo.com)
Ana Almeida (Peter_senado2002@yahoo.com)
Ilídio Jesus (Peter_senado2002@yahoo.com)

Version: 2 Date: 2 February 2013

Author's response to reviews: see over
Dear Editor,

My name is Pedro Sousa. I work in Cardiology Department in Faro Hospital, Portugal.

I think that this is an interesting case report because it highlights the unusual course of an uncorrected Tetralogy of Fallot in a 72-year-old woman. There are only a few case reports of patients with an uncorrected Tetralogy of Fallot who reach such an advanced age and for our knowledge is the first case report of a possible endocarditis in a patient with an uncorrected Tetralogy of Fallot older than 70 years.

This case also highlights the importance of transthoracic echocardiography and cardiac magnetic resonance imaging in this clinical situation.

I hope that you agree with me!

Best regards

Contact:

Dr. Pedro Sousa
Cardiology Department, Faro Hospital
Rua Leão Penedo
8000 Faro - Portugal
Hospital phone: 00351 289 891100
Mobile phone: +351 967734126
Hospital fax: 289891112
Email: Peter_senado2002@yahoo.com